Author’s response to reviews

Title: Conceptualising the technical relationship of animal disease surveillance to intervention and mitigation as a basis for economic analysis

Authors:

Barbara Häsler (bhaesler@rvc.ac.uk)
Keith S Howe (K.S.Howe@exeter.ac.uk)
Katharina DC Stärk (kstaerk@rvc.ac.uk)

Version: 5 Date: 11 September 2011

Author’s response to reviews:

Dear editors,

The following changes have been made to the manuscript

p 9, 2nd last paragraph, last sentence: ‘are’ has been replaced by ‘can be’
p 20, 2nd paragraph, 2nd sentence: ‘at time’ is now ‘at a time’
p 21: The first sentence of the 2nd paragraph is now the last sentence of the first paragraph
Reference no 11: ‘Food hazards’ has been changed to ‘foodborne hazards’
Reference no 12: ‘The control and eradication of disease’ has been replaced by ‘Surveys’.

Figure caption and Figure 1: Because the colours used in the figure legend have been changed to black (by the journal), the figure caption needed to be modified. Both in Figure 1 and its caption, the letters A, B, C and D have been added to label the different curves.